Dear Editors:

Thank you very much for your excellent work on this manuscript. Please find uploaded the most recent (copy-edited) version. Note the following revisions and rejection of proposed changes:

1) On page 16, the following statement was mistakenly bolded and presented as a heading: ‘Presentation of findings in relation to the overall objective of the study, the theoretical perspectives, and rival explanations.’ However, this statement is actually meant to be the fourth bullet in the series of items describing the analytic steps of each case’s analysis. Therefore, I have removed the bolding and placed it as the fourth bullet in this series.

On page 18, I bolded ‘List of Abbreviations’ since all other headings were bolded.

In Table 1, the description of facilitation should not be placed as a bullet (A ‘technique by which one person makes things easier for others’ [60]. Facilitation models range from doing for others to enabling others.). Therefore, I did not accept this change.

If you have any questions related to these minor points or others, do not hesitate to contact me.

Warm Regards,
Robin Urquhart